Author's response to reviews

Title: Evaluation of acute toxicity and gastroprotective activity of curcuma purpurascens BI. rhizome against ethanol-induced gastric mucosal injury in rats

Authors:

Elham Rouhollahi (e_rouhollahi@yahoo.com)
Soheil Zorofchian Moghadamtousi (soheil.zorofchian@gmail.com)
Omer Abdalla Ahmed Hamdi (omerhamdi2001@hotmail.com)
Mehran Fadaeinasab (mehranfadaie_n@yahoo.com)
Maryam Hajrezaie (Maryam_hajrezaie@yahoo.com)
Khalijah Binti Awang (khalijah@um.edu.my)
Chung Yeng Looi (looicy@um.edu.my)
Mahmood Ameen Abdulla (ammeen@um.edu.my)
Zahurin Mohamed (zahurin@um.edu.my)

Version: 4 Date: 23 September 2014

Author's response to reviews: see over
23<sup>th</sup> September 2014

Dear Prof. Mainen Moshi,

First and foremost, we would like to express our utmost gratitude for your kind consideration on our manuscript. The comment of the editorial board has been done accordingly. We hope that these responses do meet your expectations. Please do not hesitate to contact us for further clarifications.

We look forward to a positive response from you.
Thank you,
Sincerely Yours,
Elham Rouhollahi
(Corresponding Author)

Editorial comment:
Comment 1:

Please include details in your Methods section on who formally identified the plant material used in your study.

Response:
Thank you for pointing this out. The botanical identification was made by Mr. Teo Leong Eng., Faculty of Science, University of Malaya, which was included in the revised manuscript.